Author's response to reviews

Title: The case for conducting First-in-Human (Phase 0 and Phase 1) clinical trials in Low and Middle income countries.

Authors:

Lydia Kapiriri (Kapirir@mcmaster.ca)
James V. Lavery (laveryj@smh.ca)
Peter A. Singer (Peter.Singer@mrcglobal.org)
Hassan Mshinda (hmshinda@costech.or.tz)
Lorne Babiuk (lorne.babiuk@ualberta.ca)
Abdallah S. Daar (a.daar@utoronto.ca)

Version: 4 Date: 14 October 2011

Author's response to reviews: see over
Dear Editor
We wish to re-submit the revisions to: The case for doing first-in-human (phase 0 and phase 1) clinical trials in developing countries.

Title Page: We have put the title in consistent sentence font and it is identical in the manuscript and submission system

Affiliations: We have included affiliations for each author. We have also listed the e-mails as advised.

Abstract and manuscript format: We have structured the abstract and manuscript into sections of Background, discussion and summary

References: All queried references have been edited and functioning links provided.

We have removed the text highlighter, the word count and DOI numbers

Table: We have removed the vertical lines and edited it accordingly.

Thank you
Sincerely

Abdallah S Daar  FRS (C)